Author's response to reviews

Title: Retinal nerve fiber layer thickness and cognitive ability in older people: the Lothian Birth Cohort 1936 study

Authors:

Augustinus Laude (draugustinuslaude@googlemail.com)
Gerassimos Lascaratos (glascaratos@gmail.com)
Ross D Henderson (rhender4@staffmail.ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)
Baljean Dhillon (Bal.Dhillon@luht.scot.nhs.uk)

Version: 3 Date: 3 July 2012

Author's response to reviews:

The following sentence has been added to the Methods section:

Written informed consent for participation in the study was obtained from all the participants.